Author’s response to reviews

Title: Nurse-identified Patient Care and Health Services Research Priorities in the United Arab Emirates

Authors:

Nabeel Al-Yateem (nalyateem@sharjah.ac.ae)
Muna Al-Tamimi (mtamimi@sharjah.ac.ae)
Maria Brenner (Brennerm@tcd.ie)
Hanan Altawil (tawil@sharjah.ac.ae)
Alaa Ahmad (alaa@seha.ae)
Sharon Brownie (sharo.brownie@aku.edu)
Shameran Slewa-Younan (S.Younan@westernsydney.edu.au)

Version: 2 Date: 31 Dec 2018

Author’s response to reviews:

1. Please move the limitation to the discussion and remove the "limitation" subtitle.

Response: done.

2. Table 1, please correct the table title and add round 2. Also please add all characteristics for round 2 that were presented in round 1.

Response: Title modified. ALL demographics collected in round two were added to the end of the table as requested.